Author's response to reviews

Title: Rapid resolution of acute subdural hematoma in severely head-injured child: a case report

Authors:

Jae-Young Park (bleu058@hanmail.net)
Kyung-Sub Moon (moonks@chonnam.ac.kr)
Jung-Kil Lee (herhuz@naver.com)
Kyung-Woon Jeung (neoneti@hanmail.net)

Version: 3 Date: 20 September 2012

Author's response to reviews: see over
Dear Editor-in-Chief

Sep 17, 2012

We are enclosing the manuscript entitled, “Rapid resolution of acute subdural hematoma in severely head-injured child: a case report.” for consideration of publication in Journal of Medical Case Reports.

In this article, we report an unexpected infrequent case and discuss the possible mechanisms with obvious imaging studies.

The material contained in the manuscript has not been published previously, nor is it under consideration for publication elsewhere. All authors have read and approved the submission of this manuscript.

Thank you for your kind considerations in advance and I am looking forward to hearing a positive reply from you soon.

Sincerely yours,

Kyung-Sub Moon, M.D.
Associate Professor
Address: Department of Neurosurgery, Chonnam National University Hwasun Hospital,
322 Seoyang-ro, Hwasun-eup, Hwasun-gun,
Jeollanam-do, 519-763, South Korea
Tel: 82-61-379-7666
FAX: 82-61-379-7673
e-mail: moonks@chonnam.ac.kr